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BRAC Cleanup Team 
(HTW BCT) Meeting Minutes 

April 14, 2010 
 
Gail Youngblood, Fort Ord BRAC Derek, Lieberman, Ahtna 
David Eisen, USACE Bill Mabey, TechLaw 
Grant Himebaugh, RWQCB  
Franklin Mark, DTSC 

Ed Ticken, MACTEC 
Steve Sterling, DTSC (phone) 

Bill Collins, Fort Ord BRAC  Roy Evans, HGL (phone)  
David Kelly, Shaw      Peter Kelsall, Shaw   
Lewis Mitani, EPA      Melissa Broadston, Fort Ord BRAC    
Martin Hausladen, EPA     Rob Robinson, Fort Ord BRAC 
Jen Moser, Shaw      Marc Edwards, USACE 
Fred Hart, USACE      Roman Racca, DTSC 
Jeff Fenton, MACTEC (phone)    Chris Goddard, USACE 
             
Agenda  
Community Involvement Status Update 
OU1 Groundwater Remediation - HGL Status Update 
OU1 Off-Site Status Update 
OU2 and Sites 2/12 Treatment Systems Status Update 
Other Groundwater Issues Status Update 
OUCTP  Status Update 
OU2 Landfill Gas Status Update 
Site 39  Remediation Status Update 
FFA Schedules Status Update 
Calendar Update 
 
Operable Unit 1 (OU 1) Groundwater Remediation – HydroGeoLogic (HGL) 
Separate minutes will be provided by HGL. 
 
Community Relations 
The Army is preparing responses to four letters from the Fort Ord Community Advisory Group 
regarding pesticides, the Operable Unit Carbon Tetrachloride Plume Remedial Action 
Completion Report, Historical Area 161, and the Offsite portion of the Operable Unit 1 plume.  
The Army is also preparing a response to the Fort Ord Environmental Justice Network comments 
on Historical Area 161.  A handout of additional actions and activities was provided. 
 
OU 1 Off-Site  
The second rebound test is complete.  The March quarterly sampling was completed March 22, 
2010.  Quarterly sampling will continue through June 2010. A carbon change will be completed 
to remove the spent carbon and parts of the treatment system will be removed from the site.  In 
September 2010, selected OU1 offsite wells will be included in the Quarterly Sampling Program.   
A handout was provided. The OU1 Explanation of Significant Differences (ESD) was submitted 
to the BCT for review and comments on 3/22/10.  Comments are due April 21, 2010. 
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Operable Unit 2 (OU2) and Sites 2/12 Treatment Systems 
OU2 – The system operated with an average flow of 579 gpm in March 2010.   
Sites 2/12 –The system operated with an average flow of 225 gpm in March 2010.   
The draft annual report was submitted March 22, 2010.  Comments are due May 21, 2010. 
A handout was provided. 
 
Other Groundwater Issues 
The March sampling is complete.  There were no access problems.  The preliminary draft annual 
report was submitted for internal review March 22, 2010.  Comments are due April 21, 2010.  
The draft version will be submitted to the BCT in May. 
 
Operable Unit Carbon Tetrachloride Plume (OUCTP) Remedial Action 
Monthly groundwater sampling was completed in March 2010 at Area 1A.  In June 2010, 
selected Area 1A wells will be included in the Quarterly Sampling Program.  Injection began 
March 2, 2010 at Area 1B. Well and pipe installations are underway at Area 1C. A new formula 
of substrate, which has proven to enhance reductive dechlorination better than the current 
formula, will be used at Area C.  System construction continues at Areas 2A and Area 2B.   
The baseline habitat monitoring for proposed monitoring wells in the FONR South Reserve will 
be conducted in April 2010. A handout was provided. The Draft Remedial Action Work Plans 
Appendix B (Upper-180Foot Aquifer) and Appendix C (Lower 180-Foot Aquifer) were issued 
March 15, 2010.  Comments are due May 20, 2010. 
 
OU2 Landfill  
The TTU system was shut down for approximately two weeks due to low methane 
concentrations.  The system will begin operating three days per week later this week.  A handout 
was provided. 
 
Site 39 Remediation 
The Army received conditional approval for the field work variance for Ranges 39/40. Pond 
monitoring is ongoing.  Remediation is complete at Ranges 17, 18, 22, 23, 27, 27A, 29, 33, 36, 
and 43.  A total of 12,000 cubic yards have been excavated and transported to the Landfill Area 
E.  Erosion control is ongoing. The BCT has approved the revised Technical Memorandum for 
HA 27.  The draft technical memorandum for HA 22 was submitted via email April 12, 2010. 
A handout was provided. 
 
FFA Schedules 
No changes necessary. 
 
Calendar Update 
The next HTW BCT is scheduled for May 12, 2010 at 1:30 PM, BRAC Conference Room. 
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